

## Administrative Record for the US Department of Energy Rocky Flats Index

## B779 - Building Cluster B779

Last updated: 10/15/2021 numerical index

| B779-A-000001   | 9/29/1997                 |
|---|---------------------------|
| Minutes of the B779 Closure Scoping meeting, September 17, 1997   |                           |
| Rocky Mountain Remediation Service (RMRS) letter transmits minutes of the Building 77 Meeting, September 17, 1997.                              | 9 (B779) Closure Scoping  |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000001.PDF  | 657 KB                    |
| B779-A-000002   | 11/17/1997                |
| 779 Cluster Decommissioning Project Waste Management Plan (WMP)   | '                         |
| Transmits the 779 Cluster Decommissioning Project Waste Management Plan (WMP) Ro Technology Site (RFETS), for review and comment; November 1997 | cky Flats Environmental   |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000002.PDF  | 2 MB                      |
|   | 0/20/4007                 |
| B779-A-000003   | 9/29/1997                 |
| B779 Cluster Closure Project HASP   |                           |
| Building 779 (B779) Cluster Closure Project Health and Safety Plan (HASP), September 2  | 29, 1997.                 |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000003.PDF  | 2 MB                      |
| B779-A-000004   | 11/4/1997                 |
| DOP for the 779 Cluster IM/IRA, October 1997  |                           |
| Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measure / Interim October 1997.   | Remedial Action (IM/IRA), |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000004.PDF  | 14 MB                     |
| D770 A 00000F   | 11/21/1007                |
| B779-A-000005   | 11/21/1997                |
| DOP for the 779 Cluster IM/IRA, December 1997   | T D. P. I. I. I.          |
| Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measure / (IM/IRA), December 1997.                                   | Interim Remedial Action   |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000005.pdf  | 14 MB                     |
| B779-A-000006   | 10/10/1997                |
| DOP for the 779 Cluster IM/IRA, October 10, 1997  |                           |
| Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measure / (IM/IRA), October 10, 1997.                                | Interim Remedial Action   |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000006.pdf  | 15 MB                     |

**B779-A-000007** 10/20/1997

B779 DOP review comment sheet, October 1997

US Department of Energy (DOE) Legal, B779 Decommissioning Operations Plan (DOP) review comment sheet, October 1997.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000007.pdf

443 KB

**B779-A-000008** 4/30/1997

Facility Disposition Working Group note submits April 1997 DOP draft

Facility Disposition Working Group note submits the April 1997 draft of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster. Subsequent B779 DOP comments are attached.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000008.pdf

12 MB

**B779-A-000009** 1/6/1997

Draft DOP for the Building 779 Complex, January 6, 1997

Draft Decommissioning Operations Plan (DOP) for the Building 779 Cluster, January 6, 1997 - This document provides the administrative structure within which the decommissioning activities will be conducted. Section 1 provides an introduction to the governing documents for decommissioning at RFETS and a summary of B 779 history, facility layout, and operations. Section 2 identifies the organizational structure, responsibilities and lines of communication, which will be used to complete the 779 Cluster decommissioning activities. Section 3 presents a description of the areas to be decommissioned and planned decontamination/decommissioning activities. Section 4, briefly describes the facility characterization process and the identified hazards within the 779 Cluster. Section 5 establishes the criteria which will be used to demonstrate the criteria which will be used to demonstrate the 779 cluster buildings have been adequately decontaminated and they are ready to be dismantled. Section 6 is a discussion of how the nuclear safety operational controls will be reduced during the decommissioning process as the hazards within the 779 cluster are reduced. Section 7 identifies the documents and controls that will be used, in conjunction with the work packages to ensure worker health and safety during decommissioning operations. Section 8 introduces the types and volumes of wastes that the project expects to generate, and how the waste will be managed when it is generated. Section 9 describes the CERCLA process that governs the project controls and execution. Section 10 establishes the quality controls and management attributes that will be implemented during the decommissioning process. Section 11 provides an overview of the security requirements for the 779 Cluster. Review comments to date are either resolved or incorporated.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000009.PDF

14 MB

**B779-A-000010** 12/18/1996

Draft DOP for the Building 779 Complex, December 18, 1996 (Comments attached)

Draft Decommissioning Operations Plan (DOP) for the Building 779 Cluster Complex, December 18, 1996 - This DOP for the B 779 Complex is the project plan which is used to describe how the Rocky Flats Cleanup Agreement (RFCA) and Decommissioning Program Plan (DPP) requirements will be implemented to complete the B 779 Cluster Decommissioning Project; various review comments attached.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000010.PDF

B779-A-000011 1/20/1997 Submits the Building 779 Cluster DOP to SSOC Submits the attached [000016] Decommissioning Operations Plan (DOP) for the Building 779 Cluster to Safe Sites of Colorado, L.L.C. (SSOC) for comment. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000011.PDF 67 KB B779-A-000012 9/14/1994 D&D Pilot Project No. 7: Building 779, Rooms 152 and 154, Draft A Submits the enclosed Task-Specific Health and Safety Plan (HASP) Environmental Restoration Programs Division Phase I, Decontamination and Decommissioning (D&D) Pilot Project No. 7, Rocky Flats Environmental Technology Site (RFETS) Building 779, rooms 152 and 154, Draft A, September 1994; HASP completes milestone No. 130020433 https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000012.PDF 1 MB B779-A-000013 1/13/1998 Milestone change for the start of Building 779 Strip-out, March 1, 1998 Regards Milestone change for the start of Building 779 Strip-out, March 1, 1998 date change will allow sufficient time for review and response to comments on the Decommissioning Operations Plan (DOP), and allow for time needed for the Colorado Department of Public Health and Environment (CDPHE) to complete approval of the document. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000013.PDF 114 KB B779-A-000014 2/3/1998 Final DOP for the B779 Cluster Submits the attached [000016] Final Decommissioning Operations Plan (DOP) for the Building 779 Cluster to be forwarded to the US Department of Energy, Rocky Flats Field Office (DOE/RFFO) and the Colorado Department of Public Health and Environment (CDPHE); responses to public comments have been included as Appendix D. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000014.PDF 80 KB B779-A-000015 1/30/1998 Proceed with training on Gloveboxes 866 and 867, B779, room 160 Documents a January 30, 1998 discussion and US Department of Energy (DOE) approval for Building 779 Closure Project to proceed with training on Gloveboxes 866 and 867, B779, room 160. Gloveboxes 866 and 867 have been identified as training gloveboxes designated for removal prior to approval of the 779 Cluster Decommissioning Operations Plan (DOP). Radiological survey results verify that these gloveboxes are not radiologically contaminated. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000015.PDF 97 KB

**B779-A-000016** 2/1/1998

DOP for the Building 779 Cluster IM/IRA, February 1998

Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measure / Interim Remedial Action (IM/IRA), February 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000016.PDF 15 MB

B779-A-000017 12/17/1997

RLCP for the 779 Cluster Project, December 17, 1997

Reconnaissance Level Characterization Plan (RLCP) for the 779 Cluster Project, December 17, 1997. Due to the change in mission of the Rocky Flats Environmental Technology Site (RFETS/Site) from the production of nuclear components to environmental cleanup and shutdown, Building 779 and its associated facilities have no identified mission after Fiscal Year 1996. The 779 Cluster is being decommissioned to reduce operating costs and to eliminate hazards within the Cluster's building. The purpose of this characterization plan is to outline the sampling requirements and methodology for characterization the 779 Cluster. The overall characterization effort identifies the type, quantity, condition and location of radioactive and hazardous materials which are, or which me be, present as residual contamination in the subject facilities. The compilation of facility information contained herein, in conjunction with the 779 Cluster Project Files established during this investigation, brings together pertinent data from various sources to serve as a practical reference for project use during the decontamination and decommissioning efforts.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000017.PDF

2 MB

**B779-A-000018** 12/19/1997

779 Cluster Decommissioning Waste Management Plan (WMP)

(B779) 779 Cluster Decommissioning Waste Management Plan (WMP) Rocky Flats Environmental Technology Site (RFETS), December 19, 1997.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000018.PDF

2 MB

**B779-A-000019** 12/17/1997

779 Cluster Decommissioning RLCR RFETS, December 17, 1997

779 Cluster Decommissioning Reconnaissance Level Characterization Report (RLCR) Rocky Flats Environmental Technology Site (RFETS), December 17, 1997.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000019.PDF

4 MB

**B779-A-000020** 1/21/1998

Memorandum submitting the attached CRSP for the 779 Cluster, January 15, 1998

Submittal of the attached Safe Sites of Colorado, L.L.C. (SSOC) / Rocky Mountain Remediation Services, L.L.C. (RMRS) Closeout Radiological Survey Plan (CRSP) for the 779 Cluster, January 15, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000020.PDF

2 MB

B779-A-000021 1/1/1998

DOP for the B779 Cluster review comments

Decommissioning Operations Plan (DOP) for the Building 779 Cluster review comments.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000021.PDF

968 KB

**B779-A-000022** 1/1/1998

Lead Compliance Plan, Building 779 Closure Project

Rocky Mountain Remediation Services, L.L.C. (RMRS) Lead Compliance Plan, Building 779 Closure Project, B779 Rocky Flats Environmental Technology Site (RFETS) January 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000022.PDF

| B779-A-000023   | 2/11/1998                    |
|---|------------------------------|
| Review of DOP for B779 Cluster, dated January 6, 1997   |                              |
| Review of the Rocky Flats Environmental Technology Site (RFETS) Draft Decommissioning Op<br>Building 779 Cluster, dated January 6, 1997                         | perations Plan (DOP) for the |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000023.PDF  | 336 KE                       |
|   |                              |
| B779-A-000024   | 3/27/1997                    |
| Review Comment Record: Draft DOP for B779 Cluster, January 15, 1997   |                              |
| Review Comment Record: Draft Decommissioning Operations Plan (DOP) for the Building 779   | Cluster, January 15, 1997.   |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000024.PDF  | 1 MB                         |
| B779-A-000025   | 12/24/1997                   |
| Transmits the completed 779 Cluster RLCR and RLCP   |                              |
| The 779 Closure Project transmits the completed Reconnaissance Level Characterization Reports Reconnaissance Level Characterization Plan (RLCP) to Kaiser-Hill. | ort (RLCR) and               |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000025.PDF  | 69 KB                        |
| B779-A-000026   | 12/24/1997                   |
| Submits the Waste Management Plan (WMP) for the 779 Cluster   | 12/2 1/1337                  |
| Submits the attached [000018] Waste Management Plan (WMP) for the 779 Cluster.  |                              |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000026.PDF  | 73 KB                        |
|   |                              |
| B779-A-000028   | 1/16/1998                    |
| Building 779 Cluster Closure Project HASP, January 16, 1998   | '                            |
| Building 779 Cluster Closure Project Health and Safety Plan (HASP), January 16, 1998.   |                              |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000028.PDF  | 2 MB                         |
|   | '                            |
| B779-A-000029   | 2/3/1998                     |
| Submits copies of the comment resolutions for the Building 779 Cluster DOP  |                              |
| Submits the attached $[000031]$ copies of the comment resolutions to the Decommissioning O the Building 779 Cluster.  | perations Plan (DOP) for     |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000029.PDF  | 76 KE                        |
| B779-A-000030   | 11/26/1996                   |
| Activities for the subject project B779   | 2-, -3, -330                 |
| Building 779 Cluster Decommissioning Project forwards a list of activities for the subject project planned to start the first week in January 1997.             | ect, B779, which are         |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000030.PDF  | 136 KB                       |

B779-A-000031 1/20/1997 Building 779 Cluster DOP submitted for comments Building 779 Decommissioning Project Decommissioning Operations Plan (DOP) for the Building 779 Cluster submitted for comments, review comment blanks attached. (See January 1997 B779 DOP under CERCLA Administrative Record (AR) IA B779-A-000009). https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000031.PDF 141 KB B779-A-000032 10/30/1996 Miscellaneous B779 Cluster Decommissioning Project Information Miscellaneous Building 779 (B779) Cluster Decommissioning Project Information - Fowards the information requested at the regulatory kickoff meeting. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000032.PDF 208 KB B779-A-000033 5/21/1996 Review comments of Building 779 Cluster Review comments of Building 779 Cluster Integrated Decontamination and Decommissioning (D&D) Schedule (ASAP PHASE II) Volume I, Draft, May 21, 1996, https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000033.PDF 412 KB B779-A-000034 1/14/1997 Internal review of the HASP for B779 Cluster Internal review of the Health and Safety Plan (HASP) for Building 779 Cluster, blank review comment sheets and B779 HASP, January 1997. https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000034.PDF 2 MB B779-A-000036 1/6/1997 Milestone contained in WAD No. 33, titled, Remove 779 Cluster Decommissioning Operations Plan (DOP) for Building 779 Cluster completes a US Department of Energy, Rocky Flats Field Office (DOE/RFFO) milestone contained in WAD No. 33, titled, Remove 779 Cluster. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000036.PDF 48 KB 2/23/1998 B779-A-000037 Building 779 Cluster Closure Project HASP, February 23, 1998 Building 779 Cluster Closure Project Health and Safety Plan (HASP), February 23, 1998. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000037.PDF 2 MB B779-A-000038 5/5/1997 Review of the DOP for the Building 779 Cluster, request for comments Review of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster, request for comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000038.PDF **75 KB** 

B779-A-000039 4/17/1997 Provides revised NEPA Compliance Material for B779 DOP Provides revised National Environmental Policy Act (NEPA) Compliance Material (Section 9.4, Environmental Issues) for Building 779 Decommissioning Operations Plan (DOP) http://www.lm.doe.gov/cercla/documents/RockyFlats docs/B779/B779-A-000039.pdf 5 MB B779-A-000040 4/8/1997 Transmits: Decommissioning of B779, requests review and comments Letter transmits the attached Environmental Checklist (EC) - Decommissioning of Building 779 Cluster (B779), requests review and comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000040.PDF 200 KB B779-A-000042 1/1/2000 Section 9.4, Environmental Impact Issues Section 9.4, Environmental Impact Issues - A mandate to incorporate National Environmental Policy Act (NEPA) values into Rocky Flats Environmental Technology Site (RFETS) decision documents is codified in the Rocky Flats Cleanup Agreement (RFCA): This section provides a description of potential environmental impacts which may be associated with the decommissioning buildings, facilities, and equipment at RFETS. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000042.PDF 787 KB B779-A-000043 1/17/1997 Comments on DOP for the Building 779 Cluster Comments on Decommissioning Operations Plan (DOP) for the Building 779 Cluster. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000043.PDF 72 KB B779-A-000044 4/28/1997 Distributes the April 1997 Building 779 Cluster Draft DOP Distributes the April 1997 Draft of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster; DOP review and comments requested. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000044.PDF 56 KB B779-A-000045 10/1/1997 Review of DOP for the Building 779 Cluster, September 1997 Review of Decommissioning Operations Plan (DOP) for the Building 779 Cluster, dated September 1997 - requests B779 DOP review and comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000045.PDF 104 KB

B779-A-000046 10/8/1997 Air Quality Review of the DOP for the B779 Cluster Air Quality Review of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster, dated September 1997; request for B779 DOP review and comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000046.PDF 198 KB B779-A-000047 10/8/1997 Compliance and performance assurance comments on DOP for the B779 Cluster Compliance and performance assurance comments on the Decommissioning Operations Plan (DOP) for the Building 779 Cluster, dated September 1997; request for B779 DOP review and comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000047.PDF 196 KB B779-A-000048 9/25/1997 DOP for the B779 IM/IRA, September 1997 Decommissioning Operations Plan (DOP) for the Building 779 Interim Measure / Interim Remedial Action (IM/IRA), September 1997 - The purpose of this document is to meet the intent and requirements of both a DOP and an IM/IRA in accordance with the Rocky Flats Cleanup Agreement (RFCA, 1996). The Proposed Action in this document is to decommission the 779 Cluster Facilities by removing the interior equipment, decontaminating the remaining structures, and dismantling the facilities. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000048.PDF 12 MB B779-A-000049 1/7/1997 Distributes Draft DOP for the B779 Cluster, January 6, 1997, for review comments Letter distributes Draft Decommissioning Operations Plan (DOP) for the Building 779 Cluster, January 6, 1997, for review comments. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000049.PDF 82 KB 12/1/1997 B779-A-000050 Independent Waste Assessment for Building 779 at RFETS Independent Waste Assessment for Building 779 at the Rocky Flats Environmental Technology Site (RFETS) Resource Conservation Recovery Act (RCRA) Facility Investigation / Remedial Investigation (RFI/RI), December 1997. https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000050.PDF 8 MB B779-A-000051 5/21/1996 Engineering/Construction/Decommissioning review comments of 779 Cluster Transmits Engineering / Construction / Decommissioning review comments of 779 Cluster Integrated Decontamination and Decommissioning (D&D) Schedule (ASAP Phase II) Volume I, Draft, May 21, 1996; (references May 8, 1996 memo)

410 KB

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000051.PDF

**B779-A-000052** 3/12/1998

779 Cluster AB Management Project (waiver request)

779 Cluster Authorization Basis Management Project: Technical justification for deviations from the format for technical safety requirements; waiver request, No. 997-98-001.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000052.PDF

357 KB

**B779-A-000054** 2/10/1998

Surface Water Monitoring Station Installation

Surface Water Monitoring Station Installation, January 29, 1997; Surface water monitoring activities performed as part of the routine Industrial Area (IA) Interim Measure / Interim Remedial Action (IM/IRA) (See November 1994 Decision Document (DD) under CERCLA Administrative Record (AR) OU08-A-000205).

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000054.PDF

63 KB

**B779-A-000055** 3/13/1998

779 Cluster Project STARR meeting minutes

779 Cluster Project Site Technical Administrative Recore Review (STARR) meeting minutes / meeting notes concurrence; Kick-off meeting, February 10, 1998 - Final B779 STARR, March 13, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000055.PDF

64 KB

**B779-A-000056** 3/11/1998

Building 779 As Low As Reasonably Achievable (ALARA) Awareness

Building 779 As Low As Reasonably Achievable (ALARA) Awareness - Addresses identified deficiencies associated with B779 employee ALARA awareness.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000056.PDF

110 KB

**B779-A-000057** 3/23/1998

Environmental / Waste Compliance Guidance

Environmental / Waste Compliance Guidance - Decommissioning transition from Resource Conservation Recovery Act (RCRA) to the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) in 779 Cluster: The enclosed Cluster-specific guidance discusses the regulatory pathway and applicable regulatory requirements that apply to the B799 Project.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000057.PDF

331 KB

**B779-A-000058** 3/24/1998

779 Cluster AB Management Project, Project, Review and Comment Record

779 Cluster Authorization Basis (AB) Management Project, Project, Review and Comment Record; documenting all comments received by US Department of Energy, Rocky Flats Field Office (DOE/RFFO) during Cross Table on the enclosed B779 Cluster Decommissioning Safety Analysis; RFFO technical direction and Kaiser-Hill Company (K-H) comments received on a Draft version of the comment record have been incorporated.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000058.PDF

**B779-A-000059**Building 779 Cluster Closure Project HASP, March 25, 1998

Building 779 Cluster Closure Project Health and Safety Plan (HASP), March 25, 1998: HASP purpose is to identify, mitigate, and control / eliminate potential safety and health hazards associated with the Closure Project; procedures and controls will be identified in this HASP that will help prevent and reduce the risk of personnel injury and / or illness and property and / or environmental damage / impacts.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000059.PDF

2 MB

**B779-A-000060** 12/29/1997

Submits the 779 Cluster RLCR

Kaiser-Hill Company, L.L.C. (K-H) submits the Reconnaissance Level Characterization Report (RLCR) for the Building 779 Cluster Project for forwarding to the regulatory agencies.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000060.PDF

80 KB

**B779-A-000062** 5/20/1998

Request for B779 Cluster DOP IM/IRA minor mod

Request for Minor Modification of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measure / Interim Remedial Action (IM/IRA). This is to request notification to the Colorado Department of Public Health and Environment (CDPHE) of a minor modification to the B779 DOP. The intent of the modification is to clarify the quantity of Gloveboxes and Hoods in Resource Conservation Recovery Act (RCRA) Unit 90.39 that will be closed during decommissioning of the 779 Cluster and to correctly identify US Environmental Protection Agency (EPA) hazardous waste numbers associated with RCRA Unit 90.43.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000062.PDF

166 KB

**B779-A-000063** 4/8/1997

March 14, 1997 EC for the decommissioning of the B779 Cluster

National Environmental Policy Act (NEPA) - March 14, 1997 Environmental Checklist (EC) for the decommissioning of the Building 779 Cluster.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000063.PDF

217 KB

**B779-A-000065** 7/16/1998

Response to CRSP comments dated July 13, 1998

Response to Closeout Radiological Survey Plan (CRSP) comments dated July 13, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000065.PDF

758 KB

**B779-A-000066** 7/16/1998

Response to CRSP comments dated June 16, 1998

Response to Closeout Radiological Survey Plan (CRSP) comments dated June 16, 1998, additional comments included.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000066.PDF

**B779-A-000067** 4/30/1998

CERTIFICATION OF RCRA CLOSURE OF UNIT 881.3B (GLOVEBOX 460) IN BUILDING 779 EPA ID No. C07890010526

The purpose of this report is to certify the RCRA closure of Unit 881 -36 (Room 222, Glovebox 460) in Building 779 at the Rocky Flats Environmental Technology Site (Site).

http://www.lm.doe.gov/cercla/documents/RockyFlats docs/B779/B779-A-000067.PDF

111 KB

**B779-A-000070** 9/28/1998

Pertinent information regarding Unit No. 90.43 Closure

Pertinent information regarding Unit No. 90.43 Closure, Resource Conservation Recovery Act (RCRA) Closures, Building 779; US Department of Energy, Rocky Flats Field Office (DOE/RFFO) April 10, 1997 correspondence to Colorado Department of Public Health and Environment (CDPHE) regarding same, attached.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000070.PDF

159 KB

**B779-A-000071** 10/12/1998

Transmits mod for B779 Cluster DOP for demolition of B729, October, 1998

Transmits the modification for the Building 779 Cluster Decommissioning Operations Plan (DOP) for the demolition of Building 729, October, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000071.PDF

747 KB

**B779-A-000072** 12/7/1998

779 Closure Project request for DOP minor mod

779 Closure Project request for Minor Modification of the Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measures / Interim Remedial Action Plan (IM/IRA). Included is a memorandum from US Department of Energy, Rocky Flats Field Office (DOE/RFFO) to Colorado Department of Public Health and Environment (CDPHE) providing notification of the minor modification. Effective date: February 6, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000072.PDF

263 KB

B779-A-000073 1/14/1999

CRSP for the 779 Cluster

Transmits the enclosed Closeout Radiological Survey Plan (CRSP) for the 779 Cluster, Revision 1 December 1998 - This CRSP was written utilizing the consultative process defined in the Rocky Flats Cleanup Agreement (RFCA), and incorporates disposition of all final review comments from the Colorado Department of Public Health and Environment (CDPHE), the US Environmental Protection Agency (EPA), and US Department of Energy, Rocky Flats Field Office (DOE/RFFO). The plan has been reviewed and approved by Kaiser-Hill Company (K-H) Implementation of the plan, which defines the methods for collecting, analyzing and documenting data, is underway.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000073.PDF

4 MB

**B779-A-000074** 3/9/1999

Submits the 779 Cluster CRSP, March 9, 1999

Submits the attached [000073] Closeout Radiological Survey Plan (CRSP) for the 779 Cluster. See the CRSP under Administrative Record (AR) B779-A-000073.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000074.PDF

**B779-A-000075** 3/26/1999

Submits CRSP for the 779 Cluster

Submits the attached [000073] Closeout Radiological Survey Plan (CRSP) for the 779 Cluster for US Department of Energy, Rocky Flats Field Office (DOE/RFFO) review and notification to the Colorado Department of Public Health and Environment (CDPHE). The revised document incorporates additional comments received from MACTEC-ERS, RFFO's Independent Verification contractor and CDPHE. The CRSP is under Administrative Record (AR) B779-A-00073.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000075.PDF

86 KB

**B779-A-000076** 5/5/1999

B729 Decommissioning Project Non-Rad Closeout Survey Report, May 3, 1999

Building 729 Decommissioning Project Non-Radiological Closeout Survey Report, Revision 0, May 3, 1999 - The purpose of this report is to summarize and present the data necessary to demonstrate that waste chemicals, Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), and Toxic Substance Control Act (TSCA) contaminants introduced into Building 729 have been removed or reduced to levels that are no longerregulated. The following are documents cited as references for this record: Asbestos Characterization Report for the 779 Cluster Project, Revision 0, October 1997; Beryllium Surface Sample Locations; Decommissioning Operations Plan for the 779 Cluster Interim Measure / Interim Remedial Action, Revision 0, February 1998; Lead / Metals in Paint Characterization for Building 779 Cluster, Revision 0, August 6, 1998; Reconnaissance Level Characterization Report for the 779 Cluster, Revision 0, December 17, 1997; Building 779 Cluster Closure Project Health and Safety Plan, Revision 10, March 22, 1999; Radiological Closeout Survey Plan for the Building 729 Cluster, Revision 2, March 1999; Radiological Closeout Survey Report for The Building 729 Cluster, Revision 0, April 1998; RFETS Analytical Services Statement of Work; 779 Cluster Decommissioning Project Waste Management Plan, Revision 1, December 1997; Procedure DA-GR01-V1-1, Analytical Services General Guidelines for Data Verification and Validation.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000076.PDF

4 MB

**B779-A-000077** 5/5/1999

Modification Addendum to the DOP for Demolition of the 779 Cluster, April 1999

Transmittal of the enclosed Rocky Flats Environmental Technology Site (RFETS) Modification Addendum to the 779 Cluster Decommissioning Operations Plan (DOP) for Demolition of the 779 Cluster, April 1999. Includes Building 779 Cluster Demolition Project Air Monitoring Plan, March 1999. The modification incorporates comments from Colorado Department of Public Health and Environment (CDPHE) and US Environmental Protection Agency (EPA) received as part of the Rocky Flats Cleanup Agreement (RFCA) consultative process. Upon final review and concurrence by US Department of Energy, Rocky Flats Field Office (DOE/RFFO), the modification should be transmitted to the CDPHE for approval.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000077.PDF

1 MB

**B779-A-000078** 6/2/1998

Requests CDPHE comments re proposed minor mod to DOP, B779 IM/IRA

Requests comments from Colorado Department of Public Health and Environment (CDPHE) regarding proposed Minor Modification to work being done pursuant to the Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measures / Interim Remedial Action Plan (IM/IRA).

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000078.PDF

| B779-A-000079  | 4/27/1999                   |
|--|-----------------------------|
| Review comments on B729 Non-Rad Closeout Survey Report, April 27, 1999   |                             |
| Review comments on the Building 729 Non-Radiological Closeout Survey Report, April 27, 1999.   |                             |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000079.PDF   | 232 KB                      |
| B779-A-000080  | F/4/1000                    |
|  | 5/4/1999                    |
| Transmits the B729 Non-Radiological Closeout Survey Report, April 27, 1999   |                             |
| Transmits the Building 729 Non-Radiological Closeout Survey Report, April 27, 1999.  |                             |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000080.PDF   | 67 KB                       |
| B779-A-000083  | 11/17/1997                  |
| Proposed resolutions for the B779 DOP  |                             |
| Faxed informal memorandum regarding open issues and proposed resolutions for the Building 779 Decomm Operations Plan (DOP).  | issioning                   |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000083.PDF   | 243 KB                      |
| B779-A-000084  | 6/28/1999                   |
| Completion for demolition of Building 729 for performance measure  |                             |
| Provides documentation of completion for demolition of Building 729 for performance measure, Building 779 Closure Project; includes Closeout Radiological Survey Report (CRSR) for B729, Volume I, April, 1999.  | Cluster                     |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000084.PDF   | 409 KB                      |
| B779-A-000085  | 6/23/1999                   |
| Forwards the enclosed proposed modification to the B779 DOP, April 1999  |                             |
| Forwards the enclosed proposed modification to the Building 779 Decommissioning Operations Plan (DOP), The purpose of this document is to describe the activities necessary to safely demolish B779, 782, and the activities in the 779 Cluster. This modification to the DOP has been generated with input from members Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD) and Environmental Protection Agency (EPA) staff. Includes B779 Cluster Demolition Project Air Monitoring Plan, | issociated<br>s of Colorado |
|  |                             |

1 MB

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000085.PDF

**B779-A-000086** 5/18/1999

Requests B779 DOP Field Modification; Includes DOE conditional agreement

Provides request for Building 779 Decommissioning Operations Plan (DOP) Field Modification, May 18, 1999 - B779 will discharge an estimated total of 510 gallons of aqueous hazardous waste to the B779 process waste system during a time frame beginning May 24, 1999 and ending July 24, 1999. Beginning with the first discharge, B779 will conduct biweekly inspections of the portions of the system used and will continue with these inspections until closure is complete. After the discharge of the hazardous waste is complete, the system will be flushed with copious amounts of non-hazardous water and a sample will be taken at a point in the system outside of the 779 Project boundaries. The sample will be analyzed only for the containment of concern and if results are below regulatory limits, the process waste system will be closed as a regulated unit. US Department of Energy, Rocky Flats Field Office (DOE/RFFO) agrees with the plan, with the exception of the inspection frequency. Daily inspections are required when the system is actively being used to store and/or transport hazardous waste.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000086.PDF

216 KB

**B779-A-000087** 5/10/1999

Facility Walkdown Report for Removal of All Gloveboxes from Building 779

Transmits the Rocky Mountain Remediation Service (RMRS) Facility Walkdown Report for Removal of All Gloveboxes from Building 779 (B779) for Performance Measure, May 10, 1999 - The purpose of this report is to document removal of the remaining gloveboxes from B779 for the Standard Performance Measure Rating Plan. Copies of the Plates (building layout) enclosed are microfilmed at the top of Jackets 67 and 68.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000087.PDF

1 MB

**B779-A-000088** 3/9/1999

Submits the 779 Cluster CRSP, March 9, 1999

Submits the attached [00073] Kaiser-Hill Company, L.L.C. (K-H) Closeout Radiological Survey Plan (CRSP) for the Building 779 (B779) Cluster, March 9, 1999 - Attached are three copies of the revised CRSP for the B779 Cluster for Rocky Flats Field Office (RFFO) review and notification to the Colorado Department of Public Health and Environment (CDPHE). The revised document incorporates comments received from MACTEC-ERS, RFFO's Independent Verification contractor.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000088.PDF

84 KB

B779-A-000089 10/12/1998

Modification to the B779 DOP

Transmits the Kaiser-Hill Company, L.L.C. (K-H) Modification to the Building 779 Decommissioning Operations Plan (DOP) for the Demolition of B729, October 12, 1998 - The modification incorporates comments from Rocky Flats Field Office (RFFO) as discussed and dispositioned on October 8, 1998. Upon review and concurrence by RFFO, the Minor Modification should be transmitted to the Colorado Department of Public Health and Environment (CDPHE) for approval.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000089.PDF

**B779-A-000090** 4/16/1999

Provides B779 CDPHE Survey approval letter

Provides Colorado Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD) Building 779 Survey, April 16, 1999 - The State has received and reviewed the Closeout Radiological Survey Plan (CRSP) for the B779 Cluster Revision 2, dated March 1999. Pursuant to Rocky Flats Cleanup Agreement (RFCA, 1996), the plan is approved.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000090.PDF

76 KB

**B779-A-000091** 5/21/1999

Authority to Proceed with Demolition

Transmits US Department of Energy, Rocky Flats Field Office (DOE/RFFO) Building 729 Authority to Proceed with Demolition, May 21, 1999 - The DOE has received the letter of concurrence from the Colorado Department of Public Health and Environment (CDPHE) concurring with the Closeout Radiological Survey (CRS) and the Non-Radiological Closeout Survey Reports for B729. This concurrence along with the closure of all open pre-start findings clears the way for the demolition of B729. Kaiser-Hill Company, L.L.C. (K-H) is authorized to proceed with this demolition.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000091.PDF

57 KB

**B779-A-000092** 11/13/1998

Constitutes approval of Modification to the B779 Cluster November 1998 DOP

The Colorado Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD), received US Department of Energy (DOE) correspondence transmitting the draft demolition plan for Building 729 in the 779 Cluster, October 20, 1998. The Division reviewed and commented on the plan, resulting in a revised plan dated November 1998. With the revisions to the plan and the finalization of the air monitoring plan, the Department fines the plan to be acceptable. This letter constitutes approval of the "Modification to the Building 779 (B779) Cluster Decommissioning Operations Plan (DOP) for Demolition of B729," dated November 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000092.PDF

121 KB

**B779-A-000093** 5/11/1999

Meeting with CDPHE regarding Building 779 (B779) issues, May 11, 1999

Transmits Rocky Flats Environmental Technology Site (RFETS) Regulatory Contact Record; Meeting with Colorado Department of Public Health and Environment (CDPHE) regarding Building 779 (B779) issues, May 11, 1999 - A meeting was held with Chris Gilbreath (CDPHE) to discuss three issues relative to B779. Attendees included Mike Conilouge, Ted Hopkins, Randy Leitner, Chris Gilbreath and Bob Cathel. The topics of discussion and meeting notes follow...

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000093.PDF

B779-A-000094 5/18/1999

Provides request for Approval for B779 DOP Field Modification

Provides Request for Approval for Building 779 Decommissioning Operations Plan (DOP) Field Modification, May 18, 1999 - B 779 will discharge an estimated total of 510 gallons of aqueous hazardous waste to the B779 process waste system during a time frame beginning May 24, 1999 and ending July 24, 1999. Beginning with the first discharge, B779 will conduct biweekly inspections of the portions of the system used and will continue with these inspections until closure is complete. After the discharge of the hazardous waste is complete, the system will be flushed with copious amounts of non-hazardous water and a sample will be taken at a point in the system outside of the 779 Project boundaries. The sample will be analyzed only for the containment of concern and if results are below regulatory limits, the process waste system will be closed as a regulated unit.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000094.PDF

103 KB

**B779-A-000095** 5/24/1999

Request for B779 DOP minor mod

Transmits the Rocky Mountain Remediation Service (RMRS) request for Minor Modification of the Decommissioning Operations Plan (DOP) for the Building 779 (B779) Cluster Interim Measures / Interim Remedial Action Plan (IM/IRA), May 24, 1999 - The purpose of this memorandum is to request notification to Colorado Department of Public Health and Environment (CDPHE) of minor modifications to the B779 Cluster DOP. Included is US Department of Energy, Rocky Flats Field Office (DOE/RFFO) notification to Colorado Department of Public Health and Environment (CDPHE) of minor modification. Effective date: February 6, 1998

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000095.PDF

2 MB

**B779-A-000096** 1/21/1999

Meeting with CDPHE re Proposed Modifications to B779 DOP, January 20, 1999

Purpose of Contact: Transmits Rocky Flats Environmental Technology Site (RFETS) Regulatory Contact Record; Meeting with Colorado Department of Public Health and Environment (CDPHE) regarding Proposed Modifications to the Building 779 (B779) Decommissioning Operations Plan (DOP), January 20, 1999 - presents concerns with proposed modifications to the B779 DOP which would omit Gloveboxes NC-7 and 106-6 (Unit 90.39) from closure requirements...

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000096.PDF

107 KB

**B779-A-000097** 8/4/1999

Request for B779 Cluster DOP IM/IRA major mod

Transmits the Rocky Mountain Remediation Service (RMRS) request for Major Modifications of the Decommissioning Operations Plan (DOP) for the Building 779 (B779) Cluster Interim Measures / Interim Remedial Action (IM/IRA), August 4, 1999 - The purpose of this memorandum is to request notification to US Department of Energy (DOE) and Colorado Department of Public Health and Environment (CDPHE) of major modifications to the B779 Cluster DOP.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000097.PDF

**B779-A-000098** 8/25/1999

Approves the enclosed B779 Cluster DOP IM/IRA major mod

Approves the enclosed Major Modifications of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measures / Interim Remedial Action (IM/IRA), August 25, 1999 - The purpose of this memorandum is to transmit the Building 729 Demolition Plan as a major modification to the B779 DOP for the B779 Cluster. The final B729 Demolition Plan was submitted and approved by the State, but was not formally transmitted as an addendum to the B779 DOP. This letter completes that action The B729 Demolition Plan will be listed as Addendum II to the B779 DOP. The final approved plan, approval letter, and revised Table of Contents (TOC) are included with this transmittal.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000098.PDF

1 MB

**B779-A-000099** 9/1/1999

Draft CRSR for Building 779, Administration Building (Admin Area)

Draft Closeout Radiological Survey Report (CRSR) for Building 779, Administration Building dated September 1999, Revision 0. Total and removable surface contamination surveys, paint/surface media samples (survey units 77909 and 77920 only), and scan surveys were performed in each Administration Building survey unit. The number/frequency of surveys/samples collected in each survey unit was based on the guidance provided in the Multi-Agency Radiation Survey and Site Investigation Manual (MARSSIM). All survey/sample results presented in this report meet the Derived Concentration Guideline Levels (DCGLs) as defined by the Closeout Radiological Survey Plan (CRSP) for the 779 Cluster. However, the building will nor be suitable for unrestricted release until the completion of the roof and roof containment surveys, the survey of the elevator (77918), the survey of the Exhaust Dust Tower interior (77920), the removal of miscellaneous systems and equipment (e.g. conduit, fire protection systems, etc.), and the approval of the associated Property Release Evaluations (PREs). This report documents the final survey results of the Administration Building portion of 779 (including the Administration building exterior). The final surveys for other portions of B779 and the B779 Cluster will be documented in separate stand-alone reports.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000099.PDF

9 MB

**B779-A-000100** 9/15/1999

Transmits the attached Draft Building 779 Admin Area CRSR

The purpose of this memorandum is to transmit the attached Draft Building 779 Closeout Radiological Survey Report (CRSR) (September 1999) for B779, Administrative Area, for Kaiser-Hill Company, L.L.C. (K-H), US Department of Energy (DOE), and Colorado Department of Public Health and Environment (CDPHE) review.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000100.PDF

199 KB

**B779-A-000101** 7/7/1999

Notification of intent to make minor mod to the B779 DOP Cluster IM/IRA

The purpose of this letter is to provide Notification of Intent to make Minor Modification to the Decommissioning Operations Plan (DOP) for the Building 779 (B779) Cluster Interim Measure / Interim Remedial Action (IM/IRA), in accordance with the Rocky Flats Cleanup Agreement (RFCA, 1996), Part 10 Paragraph 127, July 7, 1999. These minor modifications pertain to Section 8.0, Waste Management and Section 9.0, Regulatory and Environmental Considerations.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000101.PDF

B779-A-000103 10/7/1999

Regards B779 Admin Area Non-Radiological Closeout Plan and Report

Attaches [000105] copies for Building 779 Non-Radiological Closeout Plan and Building 779 Non-Radiological Closeout Report for the Administrative Area.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000103.PDF

44 KB

**B779-A-000104** 9/27/1999

B779 Cluster Closure Project, Non-Radiological Closeout Plan

Building 779 Cluster Closure Project, Non-Radiological Closeout Plan, September 27, 1999. The purpose of this plan is to ensure that sufficient characterization and remediation (as necessary) has occurred, in accordance with Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measure / Interim Remedial Action (IM/IRA) and Reconnaissance Level Characterization Report (RLCR) for non-radiological contaminants to allow demolition of the 779 Cluster. Prior to demolition of an area, the requirements of this plan will be evoked. By completing the steps identified in this plan, Rocky Flats Environmental Technology Site (RFETS) will ensure that all contaminants of Concern have been removed, or remediated to a level below regulated levels.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000104.PDF

985 KB

**B779-A-000105** 9/29/1999

B779 Cluster Closure Project, Non-Rad Closeout Report for Admin Area, Draft 0

779 Cluster Closure Project, Non-Radiological Closeout Report for Administrative Area of Building 779, Draft 0, dated September 29, 1999. The purpose of this report is to provide the information required by the 779 Cluster Closure Project Non-Radiological Closeout Plan for the Administrative Area of B779. Additionally, this document provides characterization information that supports the Rocky Flats Environmental Technology Site (RFETS) decision to demolish the B779 Administrative Area.

https://Impublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000105.pdf

7 MB

**B779-A-000106** 10/20/1999

Transmits 779 Non-Radiological Closeout Report for Admin Area, October 13, 1999

Transmits the 779 Non-Radiological Closeout Report for the Administrative Area, October 13, 1999.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000106.PDF

41 KB

**B779-A-000107** 10/13/1999

779 Cluster Closure Project, Non-Radiological Closeout Report for Admin Area

779 Cluster Closure Project, Non-Radiological Closeout Report for Administrative Area of Building 779. The purpose of this report is to provide the information required by the 779 Cluster Closure Project Non-Radiological Closeout Plan for the Administrative Area of B779. Additionally, this document provides characterization information that supports the Rocky Flats Environmental Technology Site (RFETS) decision to demolish the B779 Administrative Area.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000107.PDF

| B779-A-000108   | 7/21/1999 |  |
|---|-----------|--|
| Transmits letter regarding revision to Final Survey Report, B729; April 1999                |           |  |
| Transmittal letter regarding the revision to Final Survey Report, Building 729; April 1999. |           |  |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000108.PDF                                | 59 KB     |  |

**B779-A-000109** 4/1/1999

Final CRSR for Building 729, April 1999

Final Closeout Radiological Survey Report (CRSR) for Building 729, April 1999. Volume I consists of the 1.0 Introduction, 2.0 DCGLs, 3.0 Background, 4.0 Quality Assurance/Quality Control (QA/QC), 5.0 Survey Results, 6.0 Conclusion. Including Appendix 1, SCM/SIMS Scan Survey Overlay Maps; Appendix 2, SCM/SIMS Quality Control Charts; Appendix 3, Data Quality Assessment; Appendix 4, SCM/SIMS MDC Evaluation. Volume II consists of Attachments A-D. Survey 72901020 was conducted on April 23, 1999 by Stanley/Pilkintion as a part of the 78201 survey. Data was gathered using SRA Surface Contamination Monitor, SCM2. The Position Sensitive Proportional Counter was operating with an efficiency of 35%. The SRA Survey Information Management System was used to provide visual imaging and analysis of the survey data and to generate this report. Volume III consists Attachment E-K

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000109.PDF

3 MB

**B779-A-000110** 11/5/1999

Request to recategorize B779 as an Industrial Facility

Correspondence from Rocky Mountain Remediation Services (RMRS): Request to recategorize Building 779 as an Industrial Facility. The purpose of this correspondence is to request approval to treat each building separately and to recategorize the remaining buildings in accordance with US Department of Energy (DOE) guidelines. Furthermore, discusses the Hazard Categorization Criteria and provides technical justification for recateforizing the buildings.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000110.PDF

341 KB

B779-A-000111 11/11/1999

Transmits the enclosed changes to the Final Survey Report for Annex B, B779

This correspondence is to transmit the enclosed changes to the Final Survey Report for Annex B, Building 779. Including the copy of the letter with information regarding an error in the Surface Media Data Summary spreadsheets (JBB-012-99) Final Survey Report, October 11, 1999. Following are the responses to IVC Comments on Closeout Radiological Survey Report for B779, Annex B.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000111.PDF

370 KB

**B779-A-000112** 10/1/1999

CRSR for Building 779, Annex B

Closeout Radiological Survey Report (CRSR) for Building 779, Annex B, October 1999 changes for the Colorado Department of Public Health and Environment (CDPHE) and US Department of Energy (DOE). Also includes Attachment B, Survey Unit 77901 Data Summary; Attachment C, Survey Unit 77902 Data Summary; Attachment D, Survey Unit 77903 Data Summary; Attachment E, Survey Unit 77904 Data Summary; Attachment F, Verification of Survey Design; Attachment H, Data Quality Assessment.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000112.pdf

B779-A-000113 10/11/1999

Transmits the Final CRSP for Building 779 Cluster

The purpose of this correspondence is to transmit the Final Radiological Closeout Survey Plan (CRSP) for Building 779 Cluster, Revision 3, October 18, 1999.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000113.PDF

78 KB

**B779-A-000114** 10/18/1999

The CRSP for the 779 Cluster, Revision 3

The Closeout Radiological Survey Plan (CRSP) for the 779 Cluster, October 1999. Including the letter of Application of Surface Contamination Guidelines from Department of Energy Order 5400.5. Also includes US Department of Energy (DOE)/Rocky Flats Field Office (RFFO) letter of concurrence, dated April 23, 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000114.PDF

4 MB

**B779-A-000115** 11/4/1999

Transmits CRSR and Non-Rad Closeout Report for B779, Annex A (Rpt not attached)

Transmittal of the Closeout Radiological Survey Report (CRSR) and the Non-Radiological Closeout Report for Building 779, Annex A. This report demonstrates that the Annex A Area of B779 meets Free Release Criteria for Radiological Constituents, and that all Contaminants of Concern (COC) within the Annex A area have been identified, removed or remediated to below regulated levels. Therefore, the Annex A Area of B779 can be released for demolition.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000115.PDF

96 KB

**B779-A-000116** 11/16/1999

Page changes to the B779 CRSR

Transmittal of page changes to the Closeout Radiological Survey Report (CRSR) for Building 779 (CRSR not attached). These changes, which represent Revision 1 of the Annex B CRSR, are to correct an error in the Surface Media Data Summary spreadsheets that was discovered during the input of media data for the Administration Area CSR.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000116.PDF

163 KB

**B779-A-000117** 10/14/1999

Transmits B779 Cluster Non-Rad Closeout Plan and Report for the Admin Area

Transmits the Building 779 Cluster Non-Radiological Closeout Plan and Non-Radiological Closeout Report for the Administrative Area of B779 (Plan and Report not included). The purpose of the Cluster Non-Radiological Closeout Plan is to ensure that sufficient characterization and remediation, for non-radiological contaminants, will be conducted to allow demolition of the B779 Cluster. The purpose of the Non-Radiological Closeout Report for the Administrative Area of B779 is to provide the information required by the Non-Radiological Closeout Plan. This information demonstrates that all Contaminants of Concern (COCs) within the Administrative Area of B779 have been identified, removed or remediated to a level below regulated levels, and that the Administrative Area can be released for demolition.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000117.PDF

B779-A-000118 10/14/1999

Submits the CRSR for Building 779, Admin Area

Submits the attached [000148] Closeout Radiological Survey Report (CRSR) for Building 779, Administrative Area for US Department of Energy, Rocky Flats Field Office (DOE/RFFO) and Colorado Department of Public Health and Environment (CDPHE) review and approval. Consistent with the Rocky Flats Cleanup Agreement (RFCA) consultative process, the document incorportates input and comments received from RFFO, CDPHE, and the US Environmental Protection Agency (EPA), on the draft report submitted previously. The report demonstrates that the Administration Area of B779 meets free release criteria and is ready for demolition.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000118.PDF

85 KB

**B779-A-000119** 12/28/1999

Transmits CRSR and the Non-Radiological Closeout Report for B779, Main Building

Transmittal of the Closeout Radiological Survey Report (CRSR) and the Non-Radiological Closeout Report for Building 779, Main Building. These reports are the Rocky Flats Field Office (RFFO) and Colorado Department of Public Health and Environment (CDPHE) review and approval. The reports demonstrate that B779, Main Building, meets free release criteria for radiological constituents, and that all Contaminates of Concern (COC) within B779, Main Building, have been identified, removed or remediated to below regulated levels. Therefore, B779, Main Building, can be released for demolition.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000119.PDF

84 KB

**B779-A-000121** 10/30/1996

Miscellaneous B779 Cluster Decommissioning Project Information

Memorandum forwarding Miscellaneous Building 779 Cluster Decommissioning Project Information along with answers to questions asked at the kickoff meeting. Attachments include; map identifying the buildings within the project, table of contents for the proposed Decommissioning Operations Plan (DOP) and summary of the project's scope.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000121.PDF

257 KB

**B779-A-000122** 1/6/1997

Transmits the 779 Cluster DOP

This letter is to transmit the Decommissioning Operations Plan (DOP) for 779 Cluster. Attachment was not enclosed.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000122.PDF

58 KB

B779-A-000123 1/7/1997

Distributes Draft DOP for the B779 Cluster, January 6, 1997, for internal review

The purpose of this letter is to distribute the Draft Decommissioning Operations Plan (DOP) for Building 779 Cluster for internal review and comment.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000123.PDF

79 KB

**B779-A-000124** 1/6/1997

Draft DOP for Building 779 Complex

Draft Decommissioning Operations Plan (DOP) for the Building 779 Complex dated January 6, 1997.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000124.PDF

**B779-A-000125** 4/28/1997

Transmits the revised 779 Cluster Decommissioning Project's DOP

The purpose of this letter is to transmit the revised 779 Cluster Decommissioning Project's Decommissioning Operations Plan (DOP) to the appropriate personnel. Copy of the DOP was not attached to this letter.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000125.PDF

234 KB

**B779-A-000126** 10/1/1997

DOP for the B779 Cluster IM/IRA, October 1997

Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measures / Interim Remedial Action (IM/IRA), October 1997.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000126.PDF

16 MB

**B779-A-000127** 4/4/1999

Regarding the approval of 779 DOP

Email regarding latest approved 779 Decommissioning Operations Plan (DOP) modification to free up the shipment of the waste packages being held. The supporting e-mail letter is attached.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000127.PDF

123 KB

**B779-A-000128** 11/11/1999

Transmittal of the CRSP for the 779 Cluster, Revision 3

Letter transmitting the Closeout Radiological Survey Plan (CRSP) for the 779 Cluster, Revision 3. This revised document incorporates the listed recommended changes which were reviewed and discussed with the Rocky Flats Field Office (RFFO), the Colorado Department of Public Health and Environment (CDPHE) and the US Environmental Protection Agency (EPA) as part of the Rocky Flats Cleanup Agreement (RFCA) consultative process. A copy of the Closeout Radiological Survey Plan was not enclosed with this transmittal.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000128.PDF

115 KB

**B779-A-000129** 1/12/2000

Completion Documents for Demolition of Building 779

Letter transmitting the enclosed Completion Documents for Demolition of Building 779. Demolition of Building 779 is complete. Radioactively contaminated waste has been packaged and stored in a compliant manner. See Attachment 1 for Waste Item Description Codes (IDC) and shipment dates. Attachment 2 contains the Certificate of Destruction for Building 779.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000129.PDF

**B779-A-000130** 9/14/1994

Request for review of Phase I HASP for D&D Pilot Project No. 7, Draft A

Interoffice correspondence: EG&G Rocky Flats, Inc. submittal of the Task-Specific Health and Safety Plan (HASP) Environmental Restoration (ER) Programs Division, Phase I, Decontamination and Decommissioning (D&D) Pilot Project No. 7, Rocky Flats Environmental Technology Site (RFETS), Building 779, Rooms 152 and 154. Revision 0, Draft A, September 2, 1994. The purpose of this task-specfic HASP is to identify personal protection standards and mandatory safety practices and procedures for personnel involved in the Decontamination and Decommissioning (D&D) project. This HASP is based on requirements for health and safety plan content established by the Occupational Safety and Health Administration (OSHA) in Title 29 Code of Federal Regulations (CFR) 1910.120. This HASP is also based on requirements from the Health and Safety Practices Manual, the 5400 series of Department of Energy Orders, and 10 CFR 835. It is requested that the review process for this document is initiated. Further requested that comments be returned by September 25, 1994.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000130.PDF

1 MB

**B779-A-000131** 3/16/2000

Transmittal of Final Decommissioning Closeout Report for the Building 779 Pad

Rocky Mountain Remediation Services (RMRS) Interoffice correspondence submitting transmittal of Draft Final Decommissioning Closeout Report for the Building 779 Pad. This report is for review and comment, dated March 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000131.PDF

46 KB

**B779-A-000132** 3/1/2000

Final Decommissioning Closeout Report For Building 779 Pad

Rocky Mountain Remediation Services (RMRS) submitting Decommissioning Closeout Report for the Building 779 Pad, dated March 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000132.PDF

31 MB

**B779-A-000133** 9/29/1998

TRU Mixed Categorization Evaluation Unit 90.43 (GB 860) Closure

Safe Sites of Colorado, L.L.C. (SSOC) and Rocky Mountain Remediation Services (RMRS) 779 Closure Project submits Transuranic Mixed (TRM) Waste categorization evaluation. The Building 779 Decommissioning Operations Plan (DOP) Section 9.1.2 states: For those gloveboxes that do not meet the Low-Level Mixed (LLM) Waste categorization, an evaluation must be conducted to determine the feasibility and benefits of reducing the Plutonium (Pu) content by strip coating or other decontamination methods. An answer "yes" to each of the following questions is required for a waste stream to be eligible for decontamination.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000133.PDF

85 KB

**B779-A-000134** 12/9/1999

Management of concrete rubble from the Building 779

US Department of Energy (DOE) correspondence regarding the management of concrete rubble from Building 779 demolition. The discussion focused on monitoring of rubble stockpile runoff, sampling of building rubble, monitoring selected personnel and the air during demolition, and sampling of fine particles that accumulate following runoff events.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000134.PDF

**B779-A-000135** 2/2/2000

Transmits B779 Cluster Closure Project Non-Radiological Closeout Report

Rocky Mountain Remediation Services (RMRS) interoffice correspondence submitting formal transmittal of Revision 0 to the Building 779 Cluster Closure Project Non-Radiological Closeout Report for Building 727, 782, 783. This report contains concerns on contaminants to be removed such as Asbestos, Beryllium, also it includes Resource Conservation and Recovery Act (RCRA) Hazardous Substance Results.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000135.PDF

47 KB

B779-A-000136 2/1/2000

Building 779 Cluster Closure Project Non-Radiological Closeout Report

Rocky Mountain Remediation Services (RMRS) submitting Building 779 Cluster Closure Project Non-Radiological Closeout Report for Building 727, 782, 783, Rev 0 dated February 1, 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000136.pdf

6 MB

**B779-A-000137** 2/19/1999

Modification to the Building 779 DOP

Colorado Department of Public Health and Environment (CDPHE) correspondence regarding modification to the Building 779 Decommissioning Operations Plan (DOP) Relating to several Resource Conservation and Recovery Act (RCRA) related provisions and statements within the original 779 DOP.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000137.PDF

65 KB

**B779-A-000138** 12/1/1999

CRSR for Building 779, Main Building

Rocky Mountain Remediation Services (RMRS) submitting Closeout Radiological Survey Report (CRSR) for Building 779 Main Building, Revision 0, dated December 1999. This report contains Overview Maps, Paint/Surface Media Samples, Removable Surface Contamination Surveys, Scan Surveys.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000138.PDF

19 MB

B779-A-000139 3/10/2000

779 Cluster DOP minor mod

US Department of Energy (DOE) Rocky Flats Field Office (RFFO) correspondence submitting Minor Modification to the Decommissioning Operations Plan (DOP) for 779 Cluster. These modifications are to clarify proposed work process and correct errors and omissions.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000139.PDF

210 KB

**B779-A-000140** 2/15/2000

Request for minor mod of decommissioning for 779 Cluster

Rocky Mountain Remediation Services (RMRS) correspondence to request notification to Colorado Department of Public Health and Environment (CDPHE) of Minor Modifications to the 779 Cluster. Included is a description of the pits and sumps associated with Buildings 779, 782 and 783. Identify that these pits and sumps have contaminated groundwater infiltrate and describe the process that will be implemented to address the treatment.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000140.PDF

B779-A-000141 2/17/2000

Request for minor mod of DOP for the B779 Cluster IM/IRA

Kaiser-Hill Company, L.L.C. (K-H) Correspondence regarding Request for Minor Modification of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measure / Interim Remedial Action. This letter is to notify Colorado Department of Public Health and Environment (CDPHE) of minor modifications which include a description of the pits and sumps associated with Buildings 779, 782 and 783 Identify that these pits and sumps have contaminated groundwater infiltrate and describe the processthat will be implemented to address the treatment of the groundwater infiltrate.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000141.PDF

96 KB

**B779-A-000142** 1/1/2000

Sampling and Survey Report, January 2000

Rocky Flats (RF) Independent Verification submitting Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results, Building 779, Main Building, dated January 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised b779-a-000142.pdf

19 MB

**B779-A-000143** 12/27/1999

Sampling and Survey Report December 1999, Admin Area

Rocky Flats Environmental Technology Site (RFETS) submits Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results Building 779, Administration Building (Administrative Area), dated December 1999.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised%20B779-A-000143.pdf

20 MB

**B779-A-000144** 7/29/1999

Independent Verification SAP For Building 779

Independent Verification Sampling and Analysis Plan (SAP) for Building 779 Cluster, dated July 1999.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000144.pdf

6 MB

B779-A-000145 1/27/2000

Sampling and Survey Report, January 2000

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results Building 779 Annex A, Survey Units 779-21 and 779-23, dated January 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised b779-a-000145.pdf

22 MB

**B779-A-000146** 11/30/1999

Sampling and Survey Report Building 729, November 1999

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results for Building 729, dated November 1999.

https://Impublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000146.pdf

B779-A-000147 3/2/1998 Forwards RLCR for B779 and 886 US Department of Energy (DOE) correspondence forwarding the Reconnaissance Level Characterization Report (RLCR) for Buildings 779 and 886 for review. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000147.PDF 75 KB B779-A-000148 10/13/1999 CRSR for Building 779, Admin Area Rocky Mountain Remediation Services (RMRS) Closeout Radiological Survey Report (CRSR) for Building 779, Administration Building (Administrative Area) dated October 1999, Revision 0. This document contains overview maps, paint / surface media samples, removable surface contamination surveys. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000148.pdf 8 MB B779-A-000149 2/9/2000 CRSR Building 727, 782, 782 Rocky Mountain Remediation Services (RMRS) submits Closeout Radiological Survey Report (CRSR) for Building 727, 782, and 783, dated February 2000 https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000149.PDF 5 MB B779-A-000150 12/13/1999 B779 Cluster Closure Project December 13, 1999 Rocky Mountain Remediation Services (RMRS) submits the enclosed Building 779 Cluster Closure Project, Non-Radiological Closeout Report for the Main Building, Utility Area and Annex B, Revision 0 dated December 13, 1999. This report has been reviewed and approved. https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised b779-a-000150.pdf 18 MB B779-A-000151 7/21/1999 Transmittal of Revision to Final Survey Report Building 729, July 1999 Rocky Mountain Remediation Services (RMRS) transmittal of revision to Final Survey Report Building 729. Also included are summary changes to 729 Report. This report is Closeout Radiological Survey Report (CRSR) for Building 729, dated July 1999, Volume I. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000151.pdf 216 KB

**B779-A-000152** 7/21/1999

CRSR B729, July 1999 Volume I

Rocky Mountain Remediation Services (RMRS) submits Closeout Radiological Survey Report (CRSR) for Building 729, Revision 1, dated July 1999, Volume I. This report includes surveys consisting of Paint/Surface Media Samples, Removable Surface Contamination, Scan Surveys and Total Surface Activity Surveys.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000152.pdf 5 MB

**B779-A-000153** 7/21/1999

CRSR B729, July 1999 Volume II

Rocky Mountain Remediation Services (RMRS) submits Closeout Radiological Survey Report (CRSR) for Building 729, dated July 1999, Volume II. This report includes surveys consisting of Paint/Surface Media Samples, Removable Surface Contamination, ScanSurveys and Total Surface Activity Surveys.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000153.pdf

17 MB

**B779-A-000154** 7/21/1999

CRSR B729, July 1999 Volume III

Rocky Mountain Remediation Services (RMRS) submits Closeout Radiological Survey Report (CRSR) for Building 729, dated July 1999, Volume III. This report includes surveys consisting of Paint / Surface Media Samples, Removable Surface Contamination, Scan Surveys and Total Surface Activity Surveys.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000154.pdf

11 MB

**B779-A-000155** 11/29/1999

Sampling and Survey Report November 1999, Building 779

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results, Building 779, Annex B, dated November 1999. This report has been approved.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000155.pdf

20 MB

**B779-A-000156** 2/9/2000

Sampling Survey Report Buildings 727, 782, 783, March 2000, Volume II

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results, for Buildings 727, 782 and 783, Survey Units 727-01, 727-02, 782-01 and 782-02, Volume II, dated March 2000. This report includes Analytical Report and Summary, Sample Cross and Radiochemical Supporting Documentation.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000156.pdf

15 MB

**B779-A-000157** 2/17/2000

Sampling Survey Report Buildings 727, 782, 783, March 2000, Volume III

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results, for Buildings 727, 782, 783, Survey Units 727-01, 727-02, 782-01 and 782-02, Volume III, dated March 2000. This report includes Analytical Report and Summary, Sample Cross and Radiochemical Supporting Documentation.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000157.pdf

B779-A-000158 3/1/2000

Sampling Survey Report Buildings 727, 782, 783, March 2000, Volume I

Rocky Flats (RF) Independent Verification Sampling and Survey Report Evaluation and Interpretation of the Residual Radiological Surface Contamination Sampling Results, for Buildings 727, 782, 783, Survey Units 727-01, 727-02, 782-01 and 782-02, Volume I, dated March 2000. This report includes Smear Sampling, Surface Media Sampling and results.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000158.PDF

15 MB

**B779-A-000159** 11/19/1999

DOE policies and practices for demolition of 779

Rocky Flats (RF) Coalition of Local Governments letter regarding US Department of Energy (DOE) policies and practices for the demolition of Building 779. The Board was pleased with the key issues you addressed regarding the demolition and rubble removal Building 779. You have increase the confidence of the Community.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000159.PDF

98 KB

**B779-A-000160** 11/4/1999

Issues surrounding demolition of B779

US Department of Energy (DOE) letter summarizing results of meeting on October 28, 1999 regarding the management of rubble in the demolition surrounding B779. The Site believes the program currently in place is based on the best technical evidence available, but we are pleased to offer clarifications and modifications on water monitoring, air monitoring, volumetric sampling, worker monitoring, protection from the elements and sampling of fines.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000160.PDF

187 KB

**B779-A-000161** 6/3/1999

Building 779 notification for removal of SC-3 systems

Notification that Building 779 will remove safety-significant (SC-3) systems as part of Decontamination and Decommissioning (D&D). This Notification will shut down, Disconnect and remove as part of Building 779/782 systems, Building 729 Instrument Air Associated with Exhaust Ventilation.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000161.PDF

103 KB

**B779-A-000162** 2/17/2000

Buildings 727, 782, and 783 Final Surveys

Buildings 727, 782, and 783 Closeout Radiological Survey Report (CRSR) and Non-Radiological Closeout Report review and approval to proceed with demolition as described within the approved demolition plan.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000162.PDF

72 KB

**B779-A-000163** 12/29/1999

779 Main Building surveys

Review and approval of Closeout Radiological Survey Report (CRSR) for Building 779, Main Building and the Non-Radiological Closeout Report for the Main Building, Utility Area and Annex B, which is now suitable for demolition as described within the approved demolition plan.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000163.PDF

| B779-A-000164  | 12/29/1999                    |
|--|-------------------------------|
| Building 779, Main Building RFFO decision to proceed with demolition   |                               |
| Authorization to proceed in the demolition of Building 779, Main Building.   |                               |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000164.PDF   | 75 KB                         |
|  | 44.42.4000                    |
| B779-A-000165  | 11/12/1999                    |
| Building 779 Annex A, Final Surveys  |                               |
| Approval of final surveys and authorization to proceed with demolition plans for Building  | ·                             |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000165.PDF   | 70 KE                         |
| B779-A-000166  | 10/25/1999                    |
| Final CRSR, 779 Admin Area   |                               |
| Review, evaluation, and approval of the Closeout Radiological Survey Report (CRSR) for Building, results. (Administrative Area)  | Building 779, Administration  |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000166.PDF   | 77 KB                         |
| B779-A-000167  | 5/21/1999                     |
| Building 729 authority to proceed with demolition  | 3, ==, = 3 3                  |
| Authorization to proceed with demolition of Building 729.  |                               |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000167.PDF   | 52 KB                         |
| B779-A-000168  | 2/6/1998                      |
| B779 DOP Approval  | 2,0,1330                      |
| Acknowledgment of receipt of the Decommissioning Operations Plan (DOP) for the 779 C<br>Interim Remedial Action (IM/IRA) and CDPHE approval to initiate decommissioning activi | -                             |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000168.PDF   | 154 KE                        |
| B779-A-000169  | 4/16/1999                     |
| Building 779 Survey; CRSP approval   | ,, = 0, = 0.00                |
| Review and approval of the Closeout Radiological Survey Plan (CRSP) for the 779 Cluster  | Revision 2. dated March 1999. |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000169.PDF   | 66 KE                         |
|  | '                             |
| B779-A-000170  | 2/17/2000                     |
| Buildings 727, 782, and 783 Final Survey   |                               |
| Concurrence with conclusion to release Buildings 727, 782, and 783 from further radiolog   | gical controls.               |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000170.PDF   | 49 KB                         |

B779-A-000171 11/2/1999 CRSR for Building 779, Annex A Rocky Mountain Remediation Services (RMRS) submits Closeout Radiological Survey Report (CRSR) for Building 779, Annex A, dated November 1999; Approval attached. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000171.PDF 11 MB B779-A-000172 2/9/2000 Transmits CRSR and Non-Radiological Survey Report B727, B782 and B783 Kaiser-Hill Company, L.L.C. (K-H) transmits Closeout Radiological (CRSR) and Non-Radiological Survey Report for Building 727, 782 and 783. This report demonstrates these buildings meet free release criteria for radiological constituents, and that all Contaminants of Concern (COC) have been identified, removed or remediated to below regulated levels. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000172.PDF 79 KB B779-A-000173 11/25/1996 B779 D&D Authorization Basis (AB) Recommendations Rocky Mountain Remediation Services, L.L.C. (RMRS) submits Building 779 Decontamination and Decommissioning (D&D) Authorization Basis (AB) Recommendations. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000173.PDF 196 KB B779-A-000174 11/21/1996 B779 Modifications Requested for Turnover to RMRS Rocky Mountain Remediation Services, L.L.C. (RMRS) submits Draft of Building 779 Authorization Basis Modifications Requested in Preparation for Building Turnover to RMRS. Requests authorization of basic work priorities to accelerate these needed changes to Building 779. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000174.PDF 1 MB B779-A-000175 7/5/1994 B779 Complex Waste and Environmental Compliance Guide, Revision 0 Rocky Flats Plant (RFP) Building 779 Complex Waste and Environmental Compliance Guide, Revision 0. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000175.PDF 2 MB B779-A-000176 1/1/1997 Draft of RLCR for B779, January 1997

2 MB

Rocky Mountain Remediation Services, L.L.C. (RMRS) submits the enclosed January 1997 draft of Reconnaissance Level

Characterization Report (RLCR) for the Building 779 Cluster

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000176.PDF

B779-A-000177 5/12/1998 Request for Minor Modification DOP 779 Cluster IM/IRA Safe Sites of Colorado, L.L.C. (SSOC) and Rocky Mountain Remediation Services, L.L.C. (RMRS) request for Minor Modification of the Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measures / Interim Remedial Action (IM/IRA). https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000177.PDF 196 KB B779-A-000178 1/1/2000 NEPA Files 97-015 on Building 779 Project Submits enclosed document numbers that are in National Environmental Policy Act (NEPA) files 97-015 on the Building 779 Project. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000178.PDF 96 KB B779-A-000179 10/28/1997 Signature Authority for WSRIC Document Modifications Requests Safe Sites of Colorado, L.L.C. (SSOC) submits Signature Authority for Waste Stream Residue Identification and Characterization (WSRIC) Document Modification Requests. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000179.PDF 55 KB B779-A-000180 10/6/1997 Draft DOP For Decommissioning of B779 Cluster, September 1997 Ver 2nd Review US Department of Energy, Rocky Flats Field Office (DOE/RFFO) submits Draft Decommissioning Operations Plan (DOP) for the Decommissioning of the Building 779 Cluster, September 1997 Version, Second Review. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000180.PDF 128 KB B779-A-000181 10/20/1997 Air Compliance Guideline Documents for Building 779 Rocky Mountain Remediation Services, L.L.C. (RMRS) submits Air Compliance Guideline Documents for Building 779. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000181.PDF 3 MB B779-A-000182 10/6/1997

CDPHE comments on 779 DOP, October 6, 1997

Colorado Department of Public Health and Environment (CDPHE) comments on Building 779 Decommissioning Operations Plan (DOP), dated October 6, 1997. These comments include housecleaning action limit for Beryllium, which was developed in the sixtys, and might not be appropriate. The comments also discuss demolition plans, lead abatement plans and other significant plans.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000182.PDF

**B779-A-000183** 10/6/1997

CDPHE comments on the B779 DOP, October 6, 1997 - different version

Comments on Building 779 Decommissioning Operations Plan (DOP), dated October 6, 1997. These comments include housecleaning action limit for Beryllium, which was developed in the sixtys, and might not be appropriate. The comments also discuss demolition plans, lead abatement plans an other significant plans, different version.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000183.PDF

368 KB

**B779-A-000184** 10/6/1997

CDPHE comments on 779 DOP, October 6, 1997 - another different version

Comments on Building 779 Decommissioning Operations Plan (DOP), dated October 6, 1997. These comments include housecleaning action limit for Beryllium, which was developed in the sixtys, and might not be appropriate. The comments also discuss demolition plans, lead abatement plans an other significant plans, another different version.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000184.PDF

342 KB

**B779-A-000185** 10/10/1997

DOE comments on the 779 DOP, October 10, 1997

Comments on the Building 779 Decommissioning Operations Plan (DOP), dated October 10, 1997. These comments include a more focused discussion of reasonable alternatives. More detail has been added to Section 9.4.2 regarding the No Action Alternative.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000185.PDF

1 MB

**B779-A-000186** 10/20/1997

Fax transmitting 779 DOP Comments

Fax Transmitting Building 779 Decommissioning Operations Plan (DOP) Comments. These comments include an additional floor plan of all levels and room numbers that are associated with 3.1.2 Decommissioning Work Area Description.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000186.PDF

163 KB

**B779-A-000187** 10/22/1997

Closure of Glovebox 460 in B779 as part of RCRA Unit 881.3B

Rocky Mountain Remediation Services, L.L.C. (RMRS) submits notification to begin Closure of Glovebox 460 in Building 779 as part of Resource Conservation and Recovery Act (RCRA) Unit 881.3B, Under Revised Closure Description Document which is attached.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000187.PDF

338 KB

**B779-A-000188** 10/27/1997

Glovebox 460 in B779 Under RCRA Unit 881.3B

Comments on Glovebox 460 in Building 779 under Resource Conservation and Recovery Act (RCRA) Unit 881.3B.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000188.PDF

| B779-A-000189  | 10/28/1997                    |
|--|-------------------------------|
| Review of most recent DOP elicits comments for resolution  |                               |
| Review of most recent Decommissioning Operations Plan (DOP) elicits comments for resolu  | tion.                         |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000189.PDF   | 75 KB                         |
| B779-A-000190  | 12/17/1997                    |
| Environmental Compliance Related Activities for B779 Closure Project   |                               |
| Fax Transmittal of Environmental Compliance Related Activities for persons responsible for   | Building 779 Closure Project. |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000190.PDF   | 100 KB                        |
| B779-A-000191  | 12/10/1997                    |
| Administrative Records Program for B779 DOP  |                               |
| Comments on Administrative Records Program for Building 779 Decommissioning Operation this include maintaining a listing of all documents generated by the project, evaluating whe subject to the AR requirements, submitting AR documents to Administrative Records and ke documents have been submitted. | ther the documents are        |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000191.PDF   | 66 KB                         |
| B779-A-000192  | 10/3/1997                     |
| Comments on B779 DOP Draft, September 25, 1997   | <u>'</u>                      |
| Comments on Decommissioning Operations Plan (DOP) Draft, dated September 25, 1997.   |                               |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000192.PDF   | 324 KB                        |
| B779-A-000193  | 9/11/1997                     |
| State Scoping Meeting for B779 D&D Projects  | -, ,                          |
| Invitation to attend a State Scoping Meeting to be held on September 17, 1997 for the Build and Decommissioning (D&D) Project.   | ding 779 Decontamination      |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000193.PDF   | 225 KB                        |
| B779-A-000194  | 12/12/1996                    |
| Deferral of RCRA Units 90.37 and 90.39 in B779   | ,                             |
| Rocky Mountain Remediation Services, L.L.C. (RMRS) submits request for deferral of Resou Recovery Act (RCRA) closures for Units 90.37 and 90.39 Glovebox units in Building 779.  | rce Conservation and          |
| https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000194.PDF   | 158 KB                        |
|  |                               |

B779-A-000195 10/30/1997 Draft RLCR for B779 Cluster, October 1997 Rocky Mountain Remediation Services, L.L.C. (RMRS) submits draft of Reconnaissance Level Characterization Report (RLCR) for the Building 779 Cluster, dated October 30, 1997. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000195.PDF 3 MB B779-A-000196 9/22/1995 Tank 5 Closure Issue Submitting details concerning the issue uncovered while conducting the triple rinse required for the closure of Tank 5. Rocky Mountain Remediation Services, L.L.C. (RMRS) Permitting and Compliance has received request to close Resource Conservation and Recovery Act (RCRA) 90 day, Tank 5 in Building 779. https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000196.PDF 619 KB B779-A-000197 5/27/2000 Submits Decommissioning Closeout Report for B779 Kaiser-Hill Company, L.L.C. (K-H) submits the Decommissioning Closeout Report for the Building 779 Closure Project, April 2000. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000197.PDF 66 KB B779-A-000198 4/1/2000 Decommissioning Closeout Report for B779, April 2000 Kaiser-Hill Company, L.L.C. (K-H) submits the enclosed Decommissioning Closeout Report for the Building 779 Closure Project, dated April 2000. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000198.PDF 32 MB B779-A-000199 2/19/1999 CDPHE confirms receipt of DOE letter requesting 779 DOP Colorado Department of Public Health and Environment (CDPHE) submits letter stating the February 16, 1999 letter was received requesting a modification to the Building 779 Decommissioning Operations Plan (DOP) relating to several Resource Conservation and Recovery Act (RCRA) related provisions and statements within the original 779 DOP. The proposal has been reviewed and finds it to be acceptable. https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000199.PDF 98 KB B779-A-000200

B779-A-000200

DOE letter regarding minor mods to B779 DOP

US Department of Energy, Rocky Flats Field Office (DOE/RFFO) submits letter regarding Minor Modifications for the Building 779 Decommissioning Operations Plan (DOP) which are made to correct errors and omissions.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000200.PDF

234 KB

**B779-A-000201** 9/9/1999

DOE forwards CRSR for B779 Annex B

US Department of Energy, Rocky Flats Field Office (DOE/RFFO) letter forwarding Closeout Radiological Survey Report (CRSR) for Building 779, Annex B.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000201.PDF

116 KB

**B779-A-000202** 9/16/1999

E-mail regarding CRSR for B779, Annex B

E-mail regarding Closeout Radiological Survey Report (CRSR) for Building 779, Annex B for Regulatory Agency review meeting to answer questions and resolve small issues.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000202.PDF

46 KB

**B779-A-000203** 11/3/1999

Building 779 Cluster Closure Project, Annex A, Revision 0

Rocky Mountain Remediation Services, L.L.C. (RMRS) submits the enclosed Building 779 Closure Project, Non-Radiological Closeout Report for the Annex A, Revision 0, dated November 3, 1999. This report includes Maps, Asbestos results, Beryllium results, Resource Conservation and Recovery Act (RCRA) hazardous substances results, solids and chemicals, waste and environmental management systems area location report and B779 Glovebox Removal Checklist.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000203.pdf

13 MB

**B779-A-000204** 8/16/2000

DOE letter forwards Decommissioning Closeout Report for B779 Closure Project

US Department of Energy, Rocky Flats Field Office (DOE/RFFO) letter forwards Decommissioning Closeout Report for the Building 779 Closure Project to Colorado Department of Public Health and Environment (CDPHE). This report serves as documentation of completion of the Rocky Flats Cleanup Agreement (RFCA) Milestone FY 00-M2. No report attached.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000204.PDF

106 KB

**B779-A-000205** 1/25/2001

**B779 Transmittal Crosswalk** 

Modifications to the Building 779 Decommissioning Operations Plan (DOP) - Rocky Mountain Remediation Services, L.L.C. (RMRS) / US Department of Energy (DOE) Transmittal Crosswalk for the CERCLA Administrative Record (AR): {1} Minor Modification addends Section 9, Regulatory and Environmental considerations. Clarifies number of Resource Conservation and Recovery Act (RCRA) storage units to be closed and US Environmental Protection Agency (EPA) hazardous waste numbers. {2} Minor Modification addends Section 9 - deletes the RCRA storage units that didn't store hazardous waste. {3} Minor Modification addends Section 9 - clarifies the state of RCRA storage units and amends inspection requirements. {4} Minor Modification addends Section 9 - incorporates fluorescent bulb crushing as a Decontamination and Decommissioning (D&D) activity {5} Minor Modification addends Section 8, Waste Management, and 9 - eliminated the use of temporary units and implements storage in accordance with Applicable or Relevant and Appropriate Requirements (ARARs); corrects typographical and grammatical errors. {6} Major Modification: 729 Demolition Plan. {7} Major Modification: 779 Demolition Plan. {8} Field Modification initiated temporary use of process waste lines for transferring arsenic-contaminated waste to B776/777. {9} Minor Modification addends Section 3, Decommissioning Overview.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000205.PDF

B779-A-000206 3/10/2000

Provides written notice of DOE's intent to make a minor mod to B779 DOP

US Department of Energy (DOE) provides written notice to the Colorado Department of Public Health and Environment (CDPHE) of its intent to make a Minor Modifications to work conducted pursuant to the Decommissioning Operations Plan (DOP) for the 779 Cluster Interim Measures / Interim Remedial Action (IM/IRA). These modifications, which modify Section 3.0, Area Descriptions and Planned Activities, are made to: {1} Clarify proposed work process; and {2} correct errors and omissions.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000206.PDF

98 KB

**B779-A-000207** 7/7/1999

Provides written notice of DOE's intent to make a minor mod to the B779 DOP

Correspondence from the US Department of Energy, Rocky Flats Field Office (DOE/RFFO) to the Colorado Department of Public Health and Environment (CDPHE) notifying that Agency of DOE's intent to make Minor Modification to the Decommissioning Operations Plan (DOP) for the Building 779 Custer Interim Measures / Interim Remedial Action (IM/IRA). These modifications pertain to Section 8.0, Waste Management and Section 9.0, Regulatory and Environmental Considerations.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000207.PDF

121 KB

**B779-A-000208** 2/19/1999

State provides approval of the proposed minor mod

Pursuant to US Department of Energy, Rocky Flats Field Office's (DOE/RFFO) request for a Minor Modification to the Building 779 Decommissioning Operations Plan (DOP) relating to several Resource Conservation and Recovery Act (RCRA) related provisions and statements within the original 779 DOP. The Colorado Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD) has reviewed this proposal and finds it to be acceptable. This letter constitutes State approval of the proposed modification.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000208.PDF

71 KB

B779-A-000209 2/16/1999

Provides written notice of DOE's intent to make a minor mod to the B779 DOP

Correspondence from the US Department of Energy, Rocky Flats Field Office (DOE/RFFO) to the Colorado Department of Public Health and Environment (CDPHE) notifying that Agency of DOE's intent to make Minor Modification to the Decommissioning Operations Plan (DOP) for the Building 779 Custer Interim Measures / Interim Remedial Action (IM/IRA). These modifications are minor in nature and are made to correct errors and omissions.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000209.PDF

170 KB

**B779-A-000210** 11/3/1998

State provides approval of proposed modification to the 729 Draft Demolition Plan

Pursuant to the October 20, 1998 US Department of Energy, Rocky Flats Field Office (DOE/RFFO) letter forwarding the Draft Demolition Plan for Building 729 in the 779 Cluster. With the revisions to the plan and the finalization of the Air Monitoring Plan, the Department finds the plan to be acceptable. This letter constitutes approval of the Modification to the 779 Cluster Decommissioning Operations Plan (DOP) for Demolition of Building 729, dated November 1998.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000210.PDF

B779-A-000211 10/20/1998

DOE proposes minor mod to B779 DOP regarding demolition of B729

US Department of Energy, Rocky Flats Field Office (DOE/RFFO) forwards a proposed modification to the Decommissioning Operations Plan (DOP) for Building 779. This modification addresses plans for the demolition of B729. The modification includes discussions on the state of the building prior to demolition, environmental monitoring, work controls, and so forth. It is proposed to decontaminate B729 prior to demolition such that it meets the free release limits specified in the 779 DOP, and to dispose of demolition waste off-site.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000211.PDF

117 KB

**B779-A-000212** 6/23/1999

DOE proposes minor mod to B779 DOP

Forwards the attached [000205] Modification for Building 779 Decommissioning Operations Plan (DOP), dated February 6, 1998, it is stated that demolition plans must be submitted for Division approval and public comment prior to execution. Because the original DOP lacked sufficient detail on the demolition of the facility, US Department of Energy, Rocky Flats Field Office (DOE/RFFO) is submitting this modification to the DOP for final review concurrent with a 30-day Public Comment Period. This modification has been generated with input from members of HZMD staff and the US Environmental Protection Agency (EPA).

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000212.PDF

104 KB

**B779-A-000213** 7/27/1999

State provides approval of the proposed mod to B779 Demolition Plan

The Colorado Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD) has received and reviewed the Modification to the 779 Cluster Decommissioning Operations Plan (DOP) for Demolition of the 779 Cluster. The modification has gone through a Public Comment Period as required by the original DOP. HZMD review indicates that the plan is acceptable and this letter constitutes approval to commence work as described within the modification.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000213.PDF

90 KB

**B779-A-000214** 6/2/1998

DOE proposes minor mod to B779 DOP

Correspondence from the US Department of Energy, Rocky Flats Field Office (DOE/RFFO) to the Colorado Department of Public Health and Environment (CDPHE) notifying that Agency of DOE's intent to make Minor Modification to the Decommissioning Operations Plan (DOP) for the Building 779 Custer Interim Measures / Interim Remedial Action (IM/IRA). These modifications are made to {1} clarify proposed work process; and [2} correct errors and omission. The proposed minor mod as detailed modifies Section 9.0, Regulatory and Environmental Considerations, Table 9-1: Decommissioning the 779 Cluster.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000214.PDF

**B779-A-000215** 1/26/2001

Discussion of D&D of B779 and comments DOE

Colorado Department of Public Health and Environment (CDPHE) letter discussing the Building 779 Closeout Report, Revision 0 dated April 2000, which is found acceptable. The CDPHE commends the US Department of Energy (DOE) for making the project so successful. This was the first decommissioning of a significantly contaminated Plutonium (Pu) weapons production building to have ever occurred anywhere. The experience and knowledge for 779 has paved the way for success in decommissioning other Pu buildings at Rocky Flats Environmental Technology Site (RFETS).

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000215.PDF

114 KB

**B779-A-000216** 1/26/2001

Approval of B779 Decommissioning Closeout Report

Colorado Department of Public Health and Environment (CDPHE) has reviewed the Decommissioning Closeout Report for Building 779, Revision 0 dated April 2000, and with several minor revisions added based on consultation during the review process, finds the document acceptable.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000216.PDF

150 KB

**B779-A-000217** 10/11/1996

Transmits the DPP for approval prior to work in B779

Rocky Mountain Remediation Services, L.L.C. (RMRS) transmits the Decommissioning Program Plan (DPP) for review and approval prior to executing field decommissioning work in Building 779. The objectives of the DDP are as follows; to reduce hazards associated with facilities and prepare them for reuse, dismantlement, or demolition by removing unacceptable levels of radioactive material and chemicals, de-energizing electrical equipment, draining systems, and removing classified tooling.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000217.PDF

11 MB

**B779-A-000218** 10/1/1997

Asbestos Characterization Report for the 779 Cluster Decommissioning Project

Asbestos Characterization Report for the 779 Cluster Decommissioning Project, Revision 0, October 1, 1997. Includes Appendix A - Inspector Certifications and Appendix B - Bulk Asbestos Sample Lab / Data Table.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised B779-A-000218.pdf

5 MB

**B779-A-000219** 4/1/1999

Forwards March 29, 1999, B779 Cluster CRSP

Forwards the March 29, 1999, Building 779 Cluster Closeout Radiological Survey Plan (CRSP). The draft March 9, 1999, Plan has been revised to incorporate quality assurance and other comments provided by Colorado Department of Public Health and Environment (CDPHE) Hazardous Waste Management Division (HZMD) staff. In addition, the document incorporates additional comments from MACTEC-ERS, the rocky flats Field Office Independent Verification (IV) contractor.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000219.PDF

B779-A-000220 4/13/2000

Purpose of Contact: Notification of placement of RCRA regulated waste in a non-regulated storage area

During a review of the Waste Environmental Management System (WEMS), it was determined that a hazardous waste container (IP2) containing circuit boards had been moved to the 779 Pad. The container was moved to the Pad on April 11. The 779 Pad is not a Resource Conservation and Recovery Act (RCRA) regulated unit. Two additional IP2 containers of lead (D008) were also taken out to the Pad. Corrective actions were undertaken. The area where the IP2 containers were being stored on the 779 Pad was in full compliance with RCRA requirements by April 12.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000220.PDF

92 KB

**B779-A-000221** 10/28/1999

Forwards the CRSR and Non-Rad Survey Report for the B779 Administrative Area

Forwards the the attached [000148] Closeout Radiological Survey Report (CRSR) and Non-Radiological Survey Report for the Building 779 Administrative area. In addition to the submittal of these reports the Rocky Flats Environmental Technology Site (RFETS) is also notifying the Colorado Department of Public Health and Environment (CDPHE) of its intent to activate the Rocky Flats Cleanup Agreement (RFCA) Standard Operating Protocol for Recycling Concrete / Building Rubble (RSOP for Recycling Concrete). The resultant concrete debris will be stored onsite in accord with the RSOP.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000221.PDF

68 KB

**B779-A-000222** 5/20/1999

CDPHE Approval: CRSR and Non-Rad Closeout Survey Reports for B729

Regarding the Closeout Radiological Survey Report (CRSR) and Non-Radiological Closeout Survey Reports for Building 729, the State accepts these reports and the conclusion that the 729 structure meets the release criteria agreed to in the Decommissioning Operations Plan (DOP). Approval to begin the demolition of 729 in accordance with the previously approved plan is granted.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000222.PDF

146 KB

**B779-A-000223** 12/26/2000

Fwds revised Final Decommissioning Closeout Report 779 Closure Project, Nov 2000.

Forwards the revised Final Decommissioning Closeout Report for the 779 Closure Project, November 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000223.PDF

55 KB

B779-A-000224 8/24/1999

Approves the enclosed B779 Cluster DOP IM/IRA major mod

Approves the enclosed Major Modifications of the Decommissioning Operations Plan (DOP) for the Building 779 Cluster Interim Measures / Interim Remedial Action (IM/IRA), August 25, 1999 - The purpose of this memorandum is to transmit the Building 729 Demolition Plan as a major modification to the B779 DOP for the B779 Cluster. The final B729 Demolition Plan was submitted and approved by the State, but was not formally transmitted as an addendum to the B779 DOP. This letter completes that action The B729 Demolition Plan will be listed as Addendum II to the B779 DOP. The final approved plan, approval letter, and revised Table of Contents (TOC) are included with this transmittal.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000224.PDF

B779-A-000225 1/26/2004

RE: Permission to backfill excavations 700-7 Remediation Project

Purpose of Contact: The cooling tower slabs (784, 785, and 786) and the cooling tower pump-house (783), all northeast of the 779 slab have been removed along with associated utilities and foundation structures to a depth of 4 feet below final grade. It is understood that the Project has concurrence from the State to proceed with backfill of the associated excavations. The project understands the following conditions for the backfill: The remainder of the pipes that fed the cooling towers will need to be sealed closed by foaming; and Analytical results of water samples taken from the tanks at 783 may indicate the need for further remediation at 783.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000225.PDF

64 KB

**B779-A-000226** 3/20/2000

779 DOP modification approval

The Colorado Department of Public Health and Environment has received US Department of Energy, Rocky Flats Field Office (DOE/RFFO) letter of March 10, 2000, requesting a minor modification of the Building 779 Decommissioning Operations Plan (DOP) relating to certain information and changes to work plans for below ground sumps and pits in the779 cluster. CDPHE has reviewed the modification and discussed the related issues between member of staff. This letter constitutes approval of the modification pursuant to Paragraph 127 of RFCA. Specific monitoring and assessments of concerns regarding the sump in 779, 782, and 783 will be performed as agreed upon in a meeting between CDPHE, EPA, and site personnel on March 7, 2000.

https://Impublicsearch.lm.doe.gov/SiteDocs/B779-A-000226.PDF

82 KB

**B779-A-000227** 3/19/2004

Re final disposition of the 782 Tunnel

Purpose of Contact: Final disposition of the 782 Tunnel. The purpose of this ER Contact Record is to confim that the 700 -7 (779) has approval from the Colorado Department of Public Health and Environment (CDPHE) for the disposition of the 782 Tunnel as follows ...

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000227.PDF

103 KB

**B779-A-000228** 3/19/2004

Re backfill of excavations at the 700-7 Remediation Project

Purpose of Contact: Backfill of excavations at the 700-7 Remediation Project. This is to confirm that the 700-7 (779) Remediation Project has approval from the Colorado Department of Public Health and Environment (CDPHE) for backfill of the excavations already performed that are associated with the removal of the following 12 items ...

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000228.PDF

128 KB

**B779-A-000229** 3/22/2004

Re path forward fo rsoul remediation near 784 - 700-7 Remediation Project

Purpose of Contact: Path forward for soil remediation near 784 - 700-7 (779) Remediation Project. This is to confirm the following discussion with the Colorado Department of Public Health and Environment (CDPHE) regarding the excavation near the former 784 Cooling Tower at the northeast corner of the 700-7 Project. This discussion is only regarding the path forward, and another approval from the CDPHE will be sought following the listed actions. This discussion does not include actions associated with the Original Process Waste Lines, which is handled through the separate OPWL crew.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000229.PDF

**B779-A-000230** 3/4/2004

ER Backfill of excavations IHSS Group 700-7 (779) Remediation Project

Purpose of Contact: This ER Contact Record is to confirm that the Backfill of excavations, Individual Hazardous Substance Site IHSS Group 700-7, Remediation Project, has approval from the Colorado Department of Public Health and Environment (CDPHE).

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000230.PDF

75 KB

**B779-A-000231** 1/26/2001

Approval of B779 Decommissioning Closeout Report

Colorado Department of Public Health and Environment (CDPHE) has reviewed the Decommissioning Closeout Report for Building 779, Revision 0 dated April 2000, and with several minor revisions added based on consultation during the review process, finds the document acceptable.

https://lmpublicsearch.lm.doe.gov/SiteDocs/Revised%20B779-A-000231.pdf

2 MB

**B779-A-000232** 12/4/2000

Submits of Rev 1 page changes to the Decommissioning Closeout Report 779 Closure Project

The purpose of this correspondence it to transmit Rev 1 page changes to the subject 779 Closure Report. These pages changes incorporated Colorado Department of Public Health and Environment (CDPHE) comments to the Rev 0 Report dated April 2000, which was submitted to Rocky Flats Project Office (RFPO) on May 24, 2000.

https://lmpublicsearch.lm.doe.gov/SiteDocs/B779-A-000232.pdf

24 MB

**B779-A-INFO** 12/18/1998

See AR documents for this entire Cluster under B779

Information Only Entry: See the Administrative Record (AR) documents for this entire Cluster under B779 - The Building 779 Cluster includes: Buildings 727, 729, 779, 780, 780A, 780B, 782, 783, 784, 785, 786, and 787.